Author's response to reviews

Title: What is a pilot or feasibility study? A review of current practice and editorial policy

Authors:

Mubashir Arain (mp08ma@sheffield.ac.uk)
Michael J Campbell (m.j.campbell@sheffield.ac.uk)
Cindy L Cooper (c.l.cooper@sheffield.ac.uk)
Gillian A Lancaster (g.lancaster@lancaster.ac.uk)

Version: 6 Date: 15 July 2010

Author's response to reviews:

A minor typo and some tidying up have been done